

Missing men: World War II casualties and structural change

Christoph Eder

Working Papers in Economics and Statistics

2016-22



University of Innsbruck Working Papers in Economics and Statistics

The series is jointly edited and published by

- Department of Banking and Finance
- Department of Economics
- Department of Public Finance
- Department of Statistics

Contact address of the editor: Research platform "Empirical and Experimental Economics" University of Innsbruck Universitaetsstrasse 15 A-6020 Innsbruck Austria Tel: + 43 512 507 7171 Fax: + 43 512 507 2970 E-mail: eeecon@uibk.ac.at

The most recent version of all working papers can be downloaded at http://eeecon.uibk.ac.at/wopec/

For a list of recent papers see the backpages of this paper.

Missing Men: World War II Casualties and Structural Change

Christoph Eder*

July 20, 2016

Abstract

A shock to the sector composition of the local labor market can affect long-run economic development of a location. Because structural change ultimately shifts labor from agriculture to services, an early transition to manufacturing may hamper longrun prosperity. The identification strategy exploits military World War II (WWII) casualties in Austrian municipalities as an exogenous shock to the local labor market. WWII casualties shifted labor out of agriculture into manufacturing in the short-run, which eventually led to a differential path of structural change. In the long-run, I find a strong and robust negative effect of WWII casualties on subsequent economic output.

JEL Classification: O14, J40, N14, R11, R12.

Keywords: Structural change, local labor markets, spatial equilibrium, World War II, Austria.

^{*}Department of Public Finance, University of Innsbruck, Universitätsstr. 15/4, 6020 Innsbruck, Austria (e-mail: christoph.eder@uibk.ac.at). I thank Anke Kessler for helpful discussion and encouragement. Fernando Aragón, Michele Battisti, Martin Halla, Chris Muris, Nathan Nunn, Simon Woodcock, participants of seminars at Simon Fraser University, Johannes Kepler University Linz (lunch time seminar), the University of Vienna, the ifo Institute at the University of Munich, and various conferences and workshops provided valuable comments. Thanks for support by the Austrian Science Funds (NFN Labor Economics and the Welfare State). All remaining errors are mine.

1 Introduction

A well-known feature of economic growth is structural change: the reallocation of economic activity away from agriculture towards manufacturing and later the service sector (Kuznets, 1957; Herrendorf *et al.*, 2014). The New Economic Geography literature has theoretically worked out that production features inherent to different sectors can cause divergent development paths (Krugman, 1991; Krugman and Venables, 1995). It seems therefore natural to inquire if structural change, as a period of rapid change in sectors of production, affects the long-run distribution of economic activity in space. However, the design-based empirical literature on economic geography has only looked at shocks to the size of the population and not of the sector composition (Davis and Weinstein, 2002; Brakman *et al.*, 2004; Miguel and Roland, 2011; Schumann, 2014).¹

In this paper, I look at the long-run consequences of a shock to the sector composition of a local labor market through its effect on local structural change. Previous research has shown that workers earn higher wages and produce more output in the medium run if they switch from agriculture to manufacturing (Bauer *et al.*, 2013; Braun and Kvasnicka, 2014; Sarvimäki *et al.*, 2016). However, there is no research if these findings also hold in the long run. Since structural change ultimately shifts labor from agriculture towards the service sector, while the size of the manufacturing sector remains relatively stable, a local economy could benefit in the long run if the labor reallocation takes place at a later point in time but shifts labor directly from agriculture to the service sector. Adding the service sector to the picture is especially interesting because Desmet and Rossi-Hansberg (2009) argue that there is a clear difference between manufacturing and services in terms of agglomeration economies in the second half of the 20th century.

I argue that the short-run adjustment of a location to a labor endowment shock can have large and unexpected consequences if the path of structural change of a location depends on the initial sector composition. In the short-run, a negative labor endowment shock leads to a shift of production resources away from labor intensive agriculture to the more capital intensive manufacturing sector (Rybczynski, 1955). Over the process of structural change small sector switching costs for workers imply that they make as few sector switches as possible. Hence, the majority of labor movements take place directly from agriculture to the growing service sector. The location with the earlier negative labor endowment shock requires less labor reallocation and remains over-represented in manufacturing. If, however, a large service sector generates local positive agglomeration effects, these locations could grow faster

 $^{^{1}}$ A notable exception is Davis and Weinstein (2008) who look at specific industries within manufacturing. However, they do not look at the service sector.

and attract more economic activity in the long-run. Desmet and Rossi-Hansberg (2009) argue that for a "young" industry, in this case services, the benefits from agglomeration exceed congestion costs, while for a "mature" industry, like manufacturing, the opposite is true. A small initial overhang in service sector employment can therefore make the difference in the long-term location of economic activity.

For the empirical analysis, I exploit military WWII casualties as a natural experiment for a labor endowment shock at the municipality level in Austria. Conscription and location of deployment of Austrian men during WWII was based on birth cohort rather than social or economic background (Morawek and Neugebauer, 1989; Overmans, 1999). Thus, the relative number of casualties that a municipality suffered during the war was arguably exogenous to local development-enhancing factors.

To explore the plausibility of the identifying assumption, I show that the share of military WWII casualties of the male population (henceforth, WWII casualty rate) is uncorrelated with pre-WWII municipality characteristics and measures of education of the local population. Additionally, the WWII casualty rate can not explain pre-WWII economic development, which speaks in favor of a causal interpretation of the estimated effects.

I find that the short-run adjustment to WWII casualties at the municipality level is indeed, as predicted by the *Rybczynski-Theorem*, a shift of labor from the agricultural to the manufacturing sector. Over the next 40 years, low-casualty municipalities experienced faster growth of the service sector than similar areas with more soldiers killed during WWII. The larger agricultural sector in low-casualty municipalities therefore transformed into a larger service sector through structural change.

In the long run, I show that WWII casualties have a large negative effect on local economic output, measured by the total wage bill of a municipality. The difference in economic output comes from a larger service sector in low-casualty municipalities (extensive margin) and higher mean wage in, both, manufacturing and services (intensive margin). This composition of the total effect on economic output is consistent with local productivity spill-overs from the service sector and consequent agglomeration of economic activity. An increase of the WWII casualty rate of one percentage point reduces today's economic output, measured by the total wage bill, by about 5 percent. The magnitude of the effect suggests that the difference in economic activity built up over time through an advantage of low-casualty municipalities in attracting economic activity.

The negative long-run effects prove robust to the inclusion of various control variables. Within-district estimates rule out many political, economic, and geographic confounding factors. As a robustness check, I estimate the effect of WWII casualties in a sample of bordering municipalities and include neighbor fixed effects to rule out unobserved regional confounding factors. The unchanged negative effect on variables of economic performance in this particular setup suggest a high degree of internal validity of the estimates.

A threat to identification that I can not directly rule out, is a different size of drafted birth cohorts. Because of data limitations, I employ the size of age groups measured either imprecisely before or after WWII as control variables and check the robustness of the main results. It turns out that the inclusion of age group variables measured in the censuses 1939, 1951, and 1961 and detailed gender-specific demographic variables from 1971 does not change the estimated short- and long-run effects.

In an attempt to rule out alternative channels how WWII casualties affect today's local economic output, I look at the effect of WWII casualties on educational outcomes of the postwar population and differences in fertility and migration. In line with the main argument of the paper, there is no effect of WWII casualties on the education level of the local population. Additionally, I can also show that fertility and migration streams were not affected by WWII casualties.

Although one can never be sure, the evidence in this paper points towards a causal effect of WWII casualties on the long-run economic development that works through the path of structural change. To the best of my knowledge, this is a channel of persistence of historic events that has not been shown before in the literature. Nunn (2014) recently surveyed the broad literature on the long-term effects of historic events and addresses a wide array of historic events and mechanisms underlying persistence.

This paper is closely related to a body of work on the long-term effects of shocks to the local population. Acemoglu *et al.* (2011) show the effect of the Holocaust in Russia on subsequent economic and political development. The authors document a negative relationship between the reduction of the Jewish population and subsequent population growth, wages, and the communist vote share. The Holocaust eliminated the middle class, that consisted mostly of Jews, and contributed in this way to a long-lasting difference between Russian cities and regions. A number of contributions argue that labor shortages can be the driving force behind adjustment processes (Habakkuk, 1962; Acemoglu, 2010). Empirical research shows that labor shortages in agriculture can affect long-run economic outcomes through the land tenure system and public good provision (Dell, 2010) and increased capital intensity in agriculture (Hornbeck and Naidu, 2014). As an addition to this literature, I show that a labor endowment shock can have an effect on long-run development through labor adjustment between sectors and consequently a different path of structural change.

This research also relates to the empirical literature on labor movement between sectors. Braun and Kvasnicka (2014) show that refugees after WWII facilitated the growth of industrial production in Germany over the next 25 years. Bauer *et al.* (2013) argue that forced migrants arriving in Germany after WWII could increase their income by leaving agriculture. Sarvimäki *et al.* (2016) confirm these results for displaced Finns. However, this literature only looks at short- to medium-run effects of labor reallocation between sectors. I show that a change to the initial sector composition can have long-run effects that are, at an aggregate level, contrary to the findings of these papers.

Finally, this paper speaks to the literature that tests for multiple equilibria in the spatial distribution of economic activity by using war-time bombings as shocks to the local economy (Davis and Weinstein, 2002, 2008; Brakman *et al.*, 2004; Miguel and Roland, 2011). These papers analyze bombings of Japanese or German cities during WWII and bombings during the Vietnam War, respectively, but fail to find long-term effects on a range of outcome measures. The interpretation of my results stand in contrast to this line of research, which suggests the relative importance of human versus physical capital in the determination of the location of economic activity. This finding is in line with Waldinger (forthcoming), who shows the persistence of human capital shocks to universities before WWII, but not physical capital shocks during WWII, in the creation of scientific knowledge.

Section 2 of the paper describes the channel of persistence and derives empirical predictions. Section 3 gives an overview of the historical context of Austria and Section 4 introduces the empirical strategy, describes the data, and explores the plausibility of the identifying assumption. The empirical results are shown in Section 5, Section 6 presents robustness checks, while Section 7 inquires alternative mechanisms of persistence. Section 8 concludes.

2 Theoretical Considerations

This paper considers a new channel, how a historic event can lead to long-run consequences. For that matter, I describe the channel in this section and present empirical predictions. The mechanism describes in which way a short-term adjustment in labor allocations between sectors can generate a long-run difference through its effect on the path of structural change.

2.1 Setup

Assume an economy with three sectors (agriculture, manufacturing, and services) and two locations. Households have preferences over agricultural, manufacturing, and service sector goods such that an increase in income leads to structural change: the expenditure share on agriculture decreases while the expenditure share on services increases. Let households also have idiosyncratic location preferences (see Moretti (2011) for details), so that a wage differential between locations attracts additional households to the high-wage location.

All locations have access to the same production technologies. The labor allocation to each sector as well as the sector composition of each location is an equilibrium outcome. However, in the described setup the allocation of labor to a specific sector and location is, in general, not uniquely determined. With four assumptions, however, we can predict the effect of a temporary shock on the evolution of the local economies.

First, I assume that both locations have the same initial conditions. *Second*, let there be small sector switching costs for workers such that workers will only switch to another sector if they can avoid a wage reduction. *Third*, agriculture and manufacturing use a constant returns to scale production technology that uses labor and capital as inputs. The difference between these two sectors is that agriculture is more labor intensive than manufacturing.

Fourth, for the service sector I impose a different assumption: the size of the local service sector generates agglomeration economies for all sectors in that locations, but only in the long-run. This assumption is not entirely new to the literature as a number of papers assume increasing returns to scale in manufacturing and constant returns to scale in agriculture in two-sector models (see, for instance, Krugman (1991), Matsuyama (1992), Krugman and Venables (1995)). Desmet and Rossi-Hansberg (2009) provide evidence that benefits from agglomeration outweigh dispersion forces in the service sector, while in manufacturing the opposite is true. The authors argue that firms benefit from agglomeration after the introduction of a new general purpose technology (GPT). The last GPT in services (information and communication technology) was only introduced after 1970, while the last GPT in manufacturing (electricity) has been in use for almost 100 years already.

2.2 Adjustment to WWII Casualties

With the basic environment and the assumptions laid out, I now look at the adjustment and development of the local economies after a reduction of the population in one location due to WWII casualties. I will derive three predictions: (a) the short-term adjustment when agriculture and manufacturing are the dominant sectors; (b) the path of structural change in the local labor markets; and (c) the long-run differences due to the agglomeration economies generated by the service sector.

Without loss of generality, let there be a *high-casualty location* and a *low-casualty location*. The negative shock to the population of the high-casualty location increases the capitallabor ratio in that location and generates a comparative advantage in the capital-intensive manufacturing sector (Rybczynski, 1955). The high-casualty location therefore shifts labor into the manufacturing sector while the low-casualty location increases its labor allocation to the agricultural sector. Hence the first empirical prediction:

Prediction 1 (Short-run adjustment) In the short-run, the high-casualty location will experience a drop in agricultural employment and an increase in manufacturing employment relative to the low-casualty location.

Through the process of structural change labor reallocates from agriculture to the service sector, while the labor share in manufacturing remains relatively constant. The sector switching costs introduced in the second assumption imply that the low-casualty location, with the majority of the agricultural workers, experiences a larger transition of labor into the service sector.

Prediction 2 (Structural change) Labor adjustments between sectors over the process of economic growth are such that the low-casualty location increases the number of workers in the service sector faster than the high-casualty location.

Notice that Predictions 1 and 2 only mention differences in the labor allocation between sectors, but not in population size between locations. If there is no wage difference between locations, there is also no population movement towards one location. That, however, changes in the long-run, when agglomeration economies manifest itself. By the fourth assumption, a larger service sector in the low-casualty location increases location-specific total factor productivity and hence local wages. The increase in wages in that location will attract workers and hence leads to differences in economic activity, measured by local output, between locations. The next empirical prediction summarizes that insight.

Prediction 3 (Long-run adjustment) The larger service sector in the low-casualty location generates agglomeration economies. Hence, wages in the low-casualty location exceed wages in the high-casualty location. As a consequence the population and economic activity in the low-casualty location increase relative to the high-casualty location.

Population movement should not be taken literally in this section. Since Austrian municipalities are small geographic units, workers do not necessarily move to a municipality if they accept a job there. A more realistic option would be a redirection of commuting streams. A high-wage municipality should have higher local employment, i.e. more workers operate in firms located within the borders of the municipality, but the residential population does not necessarily increase.

3 Historical Context

3.1 World War II

The annexation of Austria through Nazi-Germany in March 1938, generally referred to as *Anschluss*, meant the temporary disappearance of Austria as a nation state. When WWII broke out in the summer of 1939, Austrian forces were already fully integrated in the German army (Morawek and Neugebauer, 1989, p.43). In June 1938 compulsory military service was introduced and 95,000 Austrian men were drafted before the outbreak of WWII. Austrian soldiers were recruited into two regional divisions, which were employed in all major attacks throughout the war side by side with German soldiers. Over time, additional cohorts were drafted at less than yearly intervals (Morawek and Neugebauer, 1989, p.43f). Few cases of refusal are known among the Austrian army (Morawek and Neugebauer, 1989, p.49).

Over the course of the war, an estimated 1.2 million Austrian men were drafted, out of which over 250,000 died (Hagspiel, 1995, p.329). More than 100,000 returned with severe injuries, and estimates of Austrian prisoners of war are between 500,000 and 600,000 (Vocelka, 2010, p.110; Hagspiel, 1995, p.329). About 24,300 civilians died due to air raids (Bukey, 2000, p.227). Nevertheless, the population of Austria increased from 6.7 million to 7 million between 1938 and 1945, mostly due to increased birth rates before and during the first years of the war (Hagspiel, 1995, p.57f) and incoming refugees from the East (Stanek, 1985, p.17f).

3.2 Structural Change

The Austrian economy underwent fundamental changes during the inter- and post-war periods. The mechanization of agriculture and the economic crisis of the inter-war period led to a large number of unemployed workers. Only after the *Anschluss* of Austria to the German Empire did the economy recover. The rearmament of the German Empire and the resulting industrialization also dominated the Austrian economy in the following years. During WWII, the economy mostly followed the German pattern and unemployment became an unknown phenomenon (Thalmann, 1954).

After WWII, Austria was split into four different occupation zones for 10 years. The main problems of the immediate after-war period were food shortages and lack of intact physical capital. While the Soviet occupation forces dismantled remaining machines and extracted economic resources, the economy in the zones of the Western Allies benefited from the occupation forces and the introduction of the *Marshall Plan*. By 1949 the economy started to recover and production increased significantly (Thalmann, 1954, p.504). In 1955,

when all four occupation forces left Austria, the boom was in full effect and the economy grew at an annual rate of about 2.5 percent for the following decades.

The most fundamental change in the post-war economy was the massive movement of labor out of agriculture and into the service sector. While almost 60 percent of the labor force were in agriculture in 1934, this number dropped to below 10 percent today. The sector share of services followed an opposite trend, while the manufacturing sector remained relatively stable with peak employment of just over 40 percent in the 1970s.

For this paper, the timing of WWII at the beginning of the large labor movements out of agriculture is of great advantage. The shock of WWII casualties is early enough in the process of structural change that I can study the effect of changes in the initial sector composition of the local labor market on long-term economic development. On the other hand, WWII is close enough to the onset of the large sectoral reallocation of labor that the initial shock does not peter out before the process of labor movement begins.

4 Empirical Strategy and Data

In this paper, I employ Austrian municipality-level data between 1900 and 2011. In this unique dataset, I combine data from 9 waves of population censuses that enables me to track the economic development of small geographic units over a century. For this research I also collected primary data on WWII casualties at the municipality level. In this section, I first present the empirical strategy and explain the measure of WWII casualties. Only then I describe the data as the specifics of the WWII casualty variable stem directly from the identification strategy. In the last part of the section I inquire if the identifying assumption is reasonable.

4.1 Empirical Strategy

I estimate a reduced-form relationship between the WWII casualty rate (as defined below) and measures of economic activity at a later point in time. In this way, I aim to estimate the long-run consequences of the loss of men in WWII on the local economy and uncover the mechanism at work.

The main estimating equation is:

$$Y_{ij}^{t} = \alpha_{1j} + \beta_{1}^{t} S_{ij}^{1934} + \mathbf{X}_{ij}^{1934} \gamma_{1} + \varepsilon_{1ij}^{t}, \qquad (1)$$

where Y_{ij}^t is the outcome variable in municipality *i*, district *j*, measured in year *t*, eg. local wages and the number of workers. S_{ij}^{1934} is the measure of WWII casualties as described

below. X_{ij}^{1934} is a set of control variables measured in 1934, α_{1j} is a district fixed effect and ε_{1ij}^t is the error term. The effect of interest is β_1^t .

 S_{ij}^{1934} is the share of military casualties in WWII of the male population in 1934, hereafter referred to as the *WWII casualty rate*. The variable is defined as:

$$S_{ij}^{1934} = \frac{\text{Military WWII casualties}_{ij}}{\text{Male population}_{ij}^{1934}}$$

Importantly, for this variable I only use military casualties. These soldiers either died in battle, as prisoners of war in a detention camp, of a disease, etc. Importantly, they did not die in their home town, but rather, as far away as the Soviet Union, France or Africa. Austrian men were drafted by birth cohort and employed alongside German soldiers in all major battles. Initially, men could get an exemption from military service if they could prove their importance in agriculture or industrial production. However, by the time the attack against the Soviet Union started in the summer of 1941, most exemptions were canceled and additional older and younger cohorts were recruited (Morawek and Neugebauer, 1989, p.46f). At that time, millions of prisoners of war had been brought into Germany and Austria to reduce labor shortages in agriculture and industrial production (Hagspiel, 1995, p.66f). Even though a later time of conscription meant a shorter period participating in the war, the probability of death did not necessarily go down (Overmans, 1999, p.244f). New conscripts had to replace fallen soldiers in high-intensity battles and the training period was significantly reduced for later conscripts. With the formation of the *Volkssturm* in September 1944 all men of age 16 to 60 years were drafted and deployed in defense against approaching Allied forces. Finally, in February 1945 the cohort of 1929 was drafted (Hagspiel, 1995, p.84f).

The variation in the WWII casualty rate comes from two sources: (a) sampling variance, that is the natural randomness in the mean share of war casualties, and (b) from a withinmunicipality correlation of location of employment. Anecdotal evidence reports that a birth cohort of a municipality was enlisted together and then assigned to various units so that soldiers knew some other members of their unit. This is consistent with the pattern of conscription described in Abramitzky *et al.* (2011).

Because soldiers were drafted based on birth cohorts and sent to the front irrespective of their home municipality, the probability of death of a soldier should be uncorrelated with municipality characteristics. This same argument is not true for all WWII casualties. Consider an Allied air raid with the goal to destroy a factory. The destruction of the factory claimed a certain number of civilian deaths. The total WWII casualties including the civilian deaths would then be correlated with the destruction of physical capital in a municipality. But since soldiers were exposed to the risk of death far away from their home municipality, their passing should not depend on the events in their home municipality.

4.2 Data

Austrian Municipalities are the lowest administrative units and have an average population of about 3,500 people. However, municipality sizes vary greatly from 53 people in Gramais in the province of Tyrol to Vienna with a population of 1.7 million. Originally established in 1849 as an administrative unit to replace the feudal system, the municipalities have always been the government agencies with the most contact to citizens. The municipalities have an elected mayor and municipal council, which is elected every 5 or 6 years, depending on the province. The responsibilities of a municipal government include land use planning, energy and water supply, provision of schools, etc. (Bauer *et al.*, 1977, p.65f).

The data for this paper come mostly from two sources. The number of dead WWII combatants was collected from war memorials in each municipality. The outcome and control variables are from various publications of *Statistik Austria*, the federal statistical agency, and its predecessors.

WWII Casualties

In this paper WWII casualties are defined as the number of men who served in the military and did not return home after the war. To the best of my knowledge, a complete death registry for soldiers of Nazi-Germany is not available with the location of residence before the war. Overmans (1999) is the most comprehensive work on casualties of German forces during WWII and reports on massive problems in the official death registries.

Given the lack of available data on war casualties, I take another avenue to compile a dataset. Many municipalities have constructed a memorial to remember their dead and missing soldiers. Most of them list the names of the dead and sometimes also date/location of birth and/or death. I use the number of names from these war memorials for the casualty data. Various websites for genealogist provide photographs and transcripts of memorials from many places in Austria.²

Two issues arise from this approach. First, I need to ensure that civilian casualties are not included on the memorials. The memorials are usually divided in fallen or dead, missing and sometimes civilian casualties. I never include the civilian casualties and do not use the data from a memorial if there is a female name among the list of dead, as this indicates that civilian casualties are included. The war memorials should be a complete list of all dead

²The websites I used are www.denkmalprojekt.org and www.kriegerdenkmal.co.at

and missing soldiers from a municipality, because relatives of non-returning soldiers had an interest in ensuring that their fathers, sons, and husbands were honored on the memorial. Often names were added to the end of the list, when missing soldiers did not return from war captivity after a number of years. In other cases names were erased when the person unexpectedly returned.³

The second issue with these data is how to link a war memorial to a municipality and avoid double counting. Austrian municipalities are subdivided into one or more localities, which can range from a cluster of houses to a city. Localities often were independent municipalities but have been merged earlier. In most cases, the largest locality has the same name as the municipality, so it is not entirely clear if a memorial addresses the municipality or the locality. There are also memorials that refer to the area that belongs to a rectorate of the church. To reduce measurement error in the WW casualty rate variable, I impose three restrictions. First, I restrict attention to municipalities whose borders have not changed since 1934, since I do not know if the war memorial corresponds to the old borders of a municipality or the one after the border change. Second, I drop municipalities with city status in 1945 since those often have several memorials with a different number of names on it. Finally, I focus on municipalities that have only one locality to ensure that the casualty data correspond to the whole municipality. These restrictions reduce the number of potential municipalities to 528. Out of the potential sample of municipalities, I managed to collect WWII casualty data for 314 or almost 60 percent.

Panel A of Table 1 describes the war casualty data. An average of 11.7 percent of the male population in 1934 died as soldiers during WWII, with a standard deviation of 3 percent. The total number of deaths ranges between 6 and 483, with a mean of 69.3 soldiers. The population size of the sample municipalities is on average 1,180 in the year 1934 and has grown to 2,200 people by 2011.

The lack of official and accurate data on military war casualties makes it likely that the explanatory variable includes measurement error. As commonly know, as long as the measurement error is uncorrelated with the error term, the estimate of β_1^t suffers from attenuation bias and I estimate a lower bound of the true effect.

Pre-War Municipality Characteristics

Control variables of pre-war municipality characteristics are mostly drawn from the printed version of the census in 1934. The census in 1934 is the only reliable source of municipality data in the inter-war period as the censuses of 1920 and 1923 were not even fully analyzed and published due to the inferior quality of the recorded data. The census of 1939 conducted

 $^{^{3}}$ In one instance a note was added to a name when the person returned in 1958 from imprisonment.

in the entire German Empire is of good quality, but used a different principle to record the residential population. In that census, drafted soldiers were accounted to the municipality of service and not to their home municipality. Since the affected young men are central to this paper, it is important that they are accounted to the residential population, to which eventually they fail to return.

The vector X_{ij}^{1934} in equation (1) include the log of the residential population, the share of the population in agriculture, and the share of Jews in the population, all measured in 1934. In addition, I control for the market status of a municipality in 1945, the first year on record of this variable. To account for the political attitude in a municipality before the war, I include the vote share of the Social Democrats and the NSDAP in the 1930 general elections, which were the last free elections before the war. Since the geography of a municipality could influence its post-war development, I also control for time-invariant factors such as the mean slope, the mean elevation, and its interaction. Panel B of Table 1 shows the descriptive statistics of these control variables. The share of Jews and the vote share of the NSDAP are on average low values, but have a few outliers.

Economic Outcomes

Panels C and D of Table 1 report the main outcome variables. Panel C lists the share of workers in agriculture, manufacturing, and services for 1951, 1981, and 2011. These three points in time give a good overview of the evolution of sector shares in the sample municipalities. The share of workers in agriculture decreased constantly, while the share in services increased. The manufacturing share followed a hump-shaped pattern. The data come from the decennial population censuses of 1951 to 2011.

The other long-term economic outcomes in Panel D should represent measures of economic activity of a municipality. The total wage bill in 2011 is the sum of all wages and salaries paid by all firms located within a municipality in 2011. The data come from tax reports of municipalities, as they levy a tax of 3 percent on all wages and salaries paid by firms (*Kommunalsteuer*). I interpret this measure as a proxy of local GDP.

The number of local workers is another measure of economic activity, as it counts the workforce that is actually employed within the borders of a municipality. This measure differs from the working population of a municipality insofar that it also includes in-commuters, but subtracts out-commuters of a municipality. The measure has the advantage that it can also be divided into sectors and measures the size of a sector. The number of firms comes from the firm censuses, which were conducted in 1973 and every 10 years since 1981. These numbers are complementary to the measure of local workers as the number of local workers in a municipality could be high due to more firms or due to larger firms.

Local wages are drawn from the Austrian Social Security Database (ASSD).⁴ The database contains all employment relationships in Austria since 1972 and records some worker and firm information as well as annualized earnings for each employment spell. I construct a local wage variable to measure the mean wage in a specific municipality from all workers aged 25 to 60 years. Because the wage information comes from social security data, the variable is top-coded. However, this restriction binds for few observations only and the mean wage of a municipality changes little if estimated by a Tobit model. For efficiency reasons, I weight the regressions with the mean wage as the dependent variable by the number of workers out of which the mean wage is estimated.

4.3 WWII Casualties as Random Events

The identification in this paper rests on the assumption that WWII casualties are as good as random. This is admittedly a strong assumption. I now explore if this assumption is reasonable. To this end I show that the WWII casualty rate is uncorrelated with observed pre-war municipality characteristics, the education level of the local population, and prewar economic development. As Abramitzky *et al.* (2011) already show for WWI casualties in French regions, the WWII casualty rate in Austrian municipalities does not appear to depend on observed or unobserved factors of an area.

The estimation of β_1^t in equation (1) relies on the presumption, that the WWII casualty rate is uncorrelated with any uncontrolled factors influencing post-war municipality development. For sake of the argument, let there be an unobserved factor Z_{ij} that is correlated with, both, S_{ij}^{1934} and ε_{1ij}^t . This would lead to omitted variable bias such that $E[\hat{\beta}_1^t] \neq \beta_1^t$. In general, it is untestable if such a factor Z_{ij} exists since it is unobserved. In two cases, however, we can get evidence that there is no such omitted factor Z_{ij} .

First, if the unobserved factor Z_{ij} is correlated with an observed pre-war municipality characteristic X_{it}^{1934} , then there should also be a correlation between the observed pre-war municipality characteristics X_{ij}^{1934} and the WWII casualty rate S_{ij}^{1934} . When I estimate equation (2),

$$S_{ij}^{1934} = \alpha_{2j} + X_{ij}^{1934} \gamma_2 + \varepsilon_{2ij},$$
 (2)

and $\gamma_2 = 0$, then this is strong evidence against the existence of such a factor Z_{ij} .

For instance, assume that highly educated soldiers were employed in offices far away from the battle fields and less educated soldiers were instead sent to the front line. Of course a municipality with a more educated population would then suffer fewer WWII casualties and

 $^{{}^{4}}$ Zweimüller *et al.* (2009) describe the dataset in detail.

likely perform better economically after WWII. The proposed check exploits that the education level of the population is likely correlated with observed municipality characteristics like the size of the agricultural sector or the urbanization of the municipality. A missing correlation between the size of the agricultural sector and the WWII casualty rate is one piece of evidence that WWII casualties are not driven by observed and unobserved factors.

Second, if the unobserved factor Z_{ij} is not only correlated with the post-war error term, ε_{1ij}^t , but also with the pre-war counterpart, $\varepsilon_{1ij}^{pre-war}$ in equation (1), then S_{ij}^{1934} should also be correlated with $\varepsilon_{1ij}^{pre-war}$ and hence $Y_{ij}^{pre-war}$. I therefore estimate equation (1) with economic outcome variables measured in 1900 on the left hand side. If $\beta_1^t = 0$, then it is unlikely that an unobserved factor Z_{ij} drives WWII casualties and post-WWII economic development, since it does not drive pre-WWII development.

Let me continuing with the previous example. If there is no correlation between the education level of the population and pre-war municipality characteristics like the size of the agricultural sector, then the earlier check would not find any evidence of an endogeneity problem. However, if the education level is important for *post-WWII* economic development, it should also have been important for *pre-WWII* economic development. The second test checks exactly that. If any economic variable measured in 1900 is significantly correlated with the WWII casualty rate, then there is likely an unobserved factor at play, since in this setting I can rule out a causal relationship.

I now use the available data to perform these two checks of the identifying assumption. In particular, I employ the check for pre-war municipality characteristics, the mean education level of a municipality, and economic variables measured in 1900.

Pre-War Municipality Characteristics

For visual inspection of the relationship between WWII casualties and pre-war municipality characteristics, a graph of correlations of the WWII casualty rate with all control variables is shown in Figure A.2. In line with the argument that the WWII casualty rate is not affected by social, geographic, or economic features of a municipality, there is no clear relationship between the WWII casualty rate and any other variable.

For the first check described above, I estimate equation (2). Although a correlation of the WWII casualty rate, S_{ij}^{1934} , with the set of observable control variables X_{ij}^{1934} does not bias the estimate of β_1^t in equation (1) because X_{ij}^{1934} is included in the regression, the results are informative whether the WWII casualty rate is actually random. If the WWII casualty rate would be significantly correlated with observed pre-war municipality characteristics, one cannot rule out an omitted factor that drives the results.

Columns (I)-(VI) of Table 2 show the results of regressions of the WWII casualty rate

on each set of observable municipality characteristics separately. District fixed effects are always included. Only the share of Jews in 1934 is separately statistically significant at the 10 percent level. In column (VII) I include all explanatory variables together. The significant relationship with the share of Jews in 1934 disappears, but the vote share of the NSDAP, the party of Hitler, in the general elections in 1930 is significant at the 5 percent level. However, this result is entirely driven by one outlying observation with a vote share of the NSDAP of about 25 percent. Excluding that observation decreases the point estimate of the NSDAP vote share by about two thirds, as shown in column (VIII).

A F-test of the estimates of all observable municipality characteristics being jointly zero in column (VII) results in a p-value of 0.197 and increases to 0.503 when the outlying observation is dropped in column (VIII). The general picture of the relationships in Table 2 suggests that WWII casualties are generally uncorrelated with pre-war municipality characteristics. This also implies that no unobserved factors that are correlated with the observed pre-WWII municipality characteristics drive the main results.

Education

Another possible source of endogeneity that I can not rule out entirely works through education. Educated soldiers might have been assigned to less lethal tasks in the armed forces and hence had a lower probability of death. That, of course, would reduce the WWII casualty rate in more educated municipalities. Since an educated workforce is a good predictor of the local income level (Gennaioli *et al.*, 2013, 2014), all post-war results could be biased. Unfortunately, no data on education is available at the municipality level before WWII.

To tackle this issue I use the post-WWII data on the education level of a sub-group of the population that finished its formal education before the outbreak of WWII. The first age-specific education data comes from the population census in 1971. The age-group of 55 to 64 years (birth cohorts 1907 to 1916) had likely finished their education by 1939.

The correlation between the WWII casualty rate and the education level of that age group is shown in Table 3. All the estimates are close to zero and statistically insignificant, suggesting that the education level of a soldier did not influence the probability of death. However, this evidence can only be suggestive since (a) the home municipality before WWII need not be the same as the location of residence after the war, and (b) if the probability of death of a soldier depends on his education level, the age-group observed in 1971 would be selected based on the education level. If one expects that a higher education level reduces the probability of death, the estimated effect of the education level of the surviving population on the WWII casualty rate is slightly biased towards zero because the highly educated would be over-represented in 1971 compared to the pre-war situation.

Pre-War Economic Development

For the second check if WWII casualties are correlated with an unobserved factor, I look at pre-war economic development. If an unobserved municipality feature drives post-war development, it is unlikely that this characteristics just emerged right around WWII or gained importance only in the post-war period. The effect of the unobserved characteristic should therefore be measurable already before WWII. For this purpose I preform a falsification test to check if the pre-war development is correlated with the WWII casualty rate. Of course such an effect can not be causal, but must stem from omitted variable bias, which then likely also drives development after WWII.

Table 4 reports the results on a number of available outcome measures. Column (I) uses the log population in 1900 as a proxy of the economic situation at that time, but fails to find a significant effect. For all other outcomes the size of the dataset is reduced because the province of Burgenland in the East was not part of Austria at that time and corresponding census data are not available. Columns (II)-(V) use the livestock of horses, cattle, pigs, and sheep, respectively, in 1900 as a proxy for agricultural productivity. None of the specifications show a correlation with the WWII casualty rate. In the last two columns the number of large land holdings and the number of factories in the municipality are used as outcome variables. For all these economic outcome variables I fail to find a correlation with the WWII casualty rate.

From the tests in this section I conclude that there is no evidence of an omitted variable that drives the WWII casualty rate and also affects the local economy. In none of the checks there is any sign that the WWII casualty rate is systematically affected by a municipalityspecific factor. Hence, any correlation of the WWII casualty rate and post-war development is likely a causal relationship.

5 Results

5.1 Short-Run Results

The short-term prediction of the theoretical considerations in Section 2 is a switch of workers out of agriculture into manufacturing due to WWII casualties (Prediction 1). Since agriculture is the labor-intensive sector compared to manufacturing, the reduction of the labor force in high-casualty municipalities generates a comparative advantage in manufacturing.

The first post-war statistics recording the size of the agricultural sector at the municipality level are from 1951. I use data from the population census and the agricultural census to test the effect of WWII casualties on the size of the agricultural sector. The regression results are presented in Table 5. Column (I) shows that WWII casualties reduced the share of workers in the agricultural sector in 1951 by 0.49 percentage points. Column (II) shows the same effect, where the difference in log population in agriculture between 1951 and 1934 is the dependent variable. While the mean number of people in agriculture dropped by 18 percent, a one percentage point higher WWII casualty rate increased that number by another percentage point. The agricultural census of 1960 split up the number of agricultural workers into family-internal and -external. The reduction in both groups is significantly different from zero.

Note that it is highly unlikely that reverse causality drives this result. One could argue that agricultural workers were less likely to be killed in battle because they were more comfortable with being outside all the time and possibly had a better skill set for survival in a war. But then we should find the same negative correlation between WWII casualties and the share of the agricultural sector before the war. However, Table 2 actually reports an insignificant positive correlation between the 1934 agricultural share and the WWII casualty rate. That correlation remains positive and insignificant even when I include the full set of control variables in column (VII). It is therefore unlikely that reverse causality drives the result, because then we would see a consistent pattern in the correlations between these variables.

5.2 Sector Adjustment

The suggested channel of persistence in this paper is the effect of WWII casualties on the evolution of sector shares in Austrian municipalities. Does the short-run effect of a smaller agricultural sector fade out quickly or does it lead to a different development path of labor allocations across sectors? For an answer to this question I estimate the effect of WWII casualties on the shares of labor in agriculture, manufacturing, and services using data from the decennial censuses of 1951–2011.

Table 6 shows the estimated effect of the WWII casualty rate on the sector shares in agriculture (Panel A), manufacturing (Panel B), and services (Panel C) for 1951–2011. Each estimate is the result of a separate regression.⁵ The reduction in the agricultural share described in the previous section can be traced until the census of 1971, but with reduced significance. After that there is no significant or meaningful effect of WWII casualties on the share of workers in agriculture. The fading out of this effect is not surprising because (a) the overall share of workers in that sector drops from 52 percent in 1951 to only 5 percent

⁵Although sector shares are correlated between sectors, a "Seemingly Unrelated Regressions" (SUR) model would not lead to efficiency gains as the regressors are the same in each equation and hence SUR is equal to OLS (Greene, 2008, p.257f).

in 2011, and (b) Prediction 2 implies that the difference in agricultural workers transfers to a difference in service sector workers.

The short-run increase in the manufacturing sector share due to WWII casualties remains surprisingly stable with only a small drop in 2011. This finding is consistent with the theoretical part in Section 2, where the relatively stable aggregate labor input in the manufacturing sector does not require adjustments over time. The stable size of the manufacturing sector is also visible in the summary statistic, where the sector shares range consistently around 30 to 40 percent.

An assumption in the discussion of the mechanism in Section 2 is that sector switches due to structural change take place between the agricultural and the service sector. So over time the negative effect of WWII casualties on the agricultural sector should transfer to a negative effect on the service sector (Prediction 2). This is exactly what the data shows in Panel C. Since the census in 1981, WWII casualties have a significant negative effect on the size of the service sector.

The effect of WWII casualties on sector shares is also economically meaningful as a one percentage point increase in the WWII casualty rate increases the share in manufacturing in 1971 by 0.6 percentage points, which is the largest estimated effect in this table. Even 60 years after the end of WWII, the local labor market is shaped by men who did not return from the war.

5.3 Long-Run Results

Of course the big question remains: Do WWII casualties matter for economic development in the long-run? In this section I look at indicators of economic activity more than 60 years after the end of WWII. As we will see, in the long-run WWII casualties matter a lot.

Total Wage Bill

Local economic activity can be approximated by several different variables. The total wage bill in 2011 measures the sum of all salaries and wages paid out by firms within the border of a municipality. I interpret this measure as a proxy for local GDP.

The results in Table 7 clearly show that WWII casualties reduce economic activity in the long-run. Column (I) reports a reduction of the total wage bill in 2011 by about 7 percent if the WWII casualty rate increases by one percentage point without any control variables or district fixed effects. Column (II) includes the full set of of control variables and the estimated effect decreases to 4.8 percent. The preferred specification with the full set of control variables and district fixed effects in column (III) shows a reduction of the total wage bill by about 5 percent. The inclusion of pre-war control variables and controlling for unobserved district characteristics does not significantly change results, which suggests little correlation of the WWII casualty rate with cofounding factors.

A simple explanation for the observed pattern is explored in column (IV). A municipality with a high WWII casualty rate had a drop in its population and slower population growth over time. Since fewer people also generate less economic activity, the result would be trivial. In column (IV) I include the population in 2011 as a control variable instead of the population in 1934 to rule out this channel of persistence. Since the effect of WWII casualties is left almost unchanged and is even significant at the 1 percent level, I rule out that a reduction of the contemporary population is the reason for the reduction in economic activity.

As a direct results of WWII, Austria was occupied by the Allied forces for 10 years. Eder and Halla (2015) present evidence that the economic development over the entire post-WWII period was different in the former Soviet occupation zone than in the zones of the Western Allies. It is therefore natural to inquire if the long-run effect of WWII casualties was different in the former Soviet occupation zone. Column (V) includes an interaction term of the WWII casualty rate and a Soviet zone indicator. The insignificant estimate of the interaction term suggests that the effect of the WWII casualty rate is not different in the former Soviet occupation zone.

The magnitude of the estimated effect is remarkable, as a one percentage point increase in the WWII casualty rate reduces the total wage bill of a municipality by about 5 percent. A reasonable explanation is that the effect built up over time through agglomeration economies. If firms are more productive in an area, they grow over time and additional firms open up. This attracts more workers and increases local economic activity. Eder and Halla (2015) also use data from Austrian municipalities and show that a population shock can lead to an even larger difference in economic activity 50 years later. Hence the relatively small shortrun shock can plausibly have generated the large long-run effect through agglomeration economies.

Intensive Margin: Local Wages

The total wage bill can be lower in high-casualty municipalities for two reasons: workers get paid less or fewer employed workers. With data from the *Austrian Social Security Database* (ASSD) I now investigate the first reason. The outcome variable is the log of mean annualized wages paid out by firms within a municipality to workers of age 25–60.

In Panel A of Table 8 the effect of WWII casualties on mean municipality wages are shown for 10-year intervals since the 1970s. WWII casualties reduced wages significantly at least since 1981. The negative effect of -1.4 percent in 2011 is large in economic terms: the average worker in a municipality with a WWII casualty rate of one standard deviation below the mean earns about 8.4 percent less than the average worker in a municipality with a WWII casualty rate of one standard deviation above the mean.

The effect of WWII casualties on different subgroups of workers are shown in Panel B of Table 8 for 2011. The effect is quite similar for males and females. The effect on local wages in agriculture in column (III) is not very meaningful, since there are few workers in an employment relationship in that sector. Only 187 of the 314 municipalities in the sample have at least one employed worker in agriculture. Hence it is not surprising, that there is no significant effect of WWII casualties on wages in that sector. Both, the manufacturing and service sector show a significant negative effect of similar size due to WWII casualties.

However, these numbers do not imply that the average worker in a high-casualty municipality is comparable to the average worker in a low-casualty municipality. The demographic information in the ASSD allow me to rule out that differences in the age structure, sex ratio, foreigners, and education level of the workers generate the effect on local wages. Although an effect of WWII casualties on wages that works through educational attainment would also be causal, in this way I can eliminate this channel of persistence.⁶

Table B.1 in the Appendix shows the effect of the WWII casualty rate of the municipality of employment on wages at the individual level. Controlling for personal characteristics like age, gender, education level, foreigner status, and a set of industry and job type indicators does not change the estimated effect much. This suggests that the effect of WWII casualties on today's economic performance does not run through an effect on the composition of the workforce like educational attainment.

Extensive Margin: Local Employment

The second reason for a higher local wage bill in low-casualty municipalities is that there are more workers employed. I test this with two different outcome measures: the number of local workers⁷ and the number of firms. The two outcome measures also enable me to explore the difference in economic activity at the sector level.

In Panel A of Table 9, the number of local workers is regressed on the WWII casualty rate, the usual municipality control variables, and district fixed effects. The effect on the total number of workers is -2.8 percent and is significantly different from zero. Columns (II) and (III) show no effect of WWII casualties on the number of workers in agriculture or manufacturing. The full difference in the number of local workers stems from the service

⁶For a detailed discussion of this potential mechanism, see Section 7.

⁷Local workers are all people who are employed in a municipality, irrespective of their municipality of residence.

sector as column (IV) shows. A one percentage point increase of the WWII casualty rate reduces the number of workers in the service sector by 3.3 percent.

In Panel B, I use the number of firms as the outcome measure and find a very similar pattern. The total number of firms is lower in high-casualty municipalities, but to a smaller extent than the number of local workers. Again the service sector is responsible for the overall difference, while the effect on the number of agricultural and manufacturing firms is statistically insignificant.

I conclude that the reduction in economic activity due to WWII casualties is robust to the outcome measure used. The total wage bill shows a large difference, which comes to about one third from a difference in the wage level and about two thirds from the number of workers. More specifically, the effect seems to be driven by differences in the size of the service sector, which is consistent with the theoretical considerations.

6 Robustness Checks

6.1 Neighboring Municipalities

I now present an additional check to support the causal interpretation of the estimated effect of WWII casualties. The robustness checks exploits the fact that neighboring municipalities share a large degree of possible confounding factors of economic development.

The credibility of the estimated effect of WWII casualties on long-run economic outcomes depends crucially on the identifying assumption that unobserved pre-war municipality characteristics are uncorrelated with the WWII casualty rate. As this assumption is not directly testable, an additional robustness check should underline that the results are not driven by endogenous factors.

For this purpose, I group municipalities into pairs that share a common municipality border and estimate the effect of WWII casualties within this pair. Since neighboring municipalities share many geographic features and enjoy the influence of external factors, such as the proximity to a city, in a similar way, a possible endogeneity problem should matter less in such a pair. The disadvantage of this approach is that the sample size decreases as I need to find pairs of neighboring municipalities for which I have data on WWII casualties. I then run a regression of the outcome variable on the WWII casualty rate, the usual control variables and a pair fixed effect for each municipality pair. These pair fixed effects account for all unobserved factors that matter for both municipalities in the same way and hence result in a high degree of internal validity of the findings.

The effect of WWII casualties on the total wage bill in 2011 is shown in Table 10. The

results are almost identical to the comparable specification in column (III) in Table 7. In column (II) I restrict the sample to bordering municipalities of the same province (Sample I), in column (III) to neighboring municipalities of the same district (Sample II), and in column (IV) to municipality pairs with the same market status in 1945 (Sample III). The estimated effect does not change much in any of these samples.

Table B.2 in the Appendix shows the same robustness exercise for the local wage level. The sample of all neighboring municipalities in column (I) shows an almost identical effect than the preferred specification in Table 8 of -1.3 percent. The restriction to municipality pairs of the same province (column (II)) or the same district (column (III)) does not change the result much. The estimated effect drops to -0.8 percent if only municipalities of the same market status in 1945 are used, but it remains significant at the 5 percent level.

This evidence suggests that unobserved external factors or geographic features are not correlated with the WWII casualty rate and hence do not drive the main results of the paper.

6.2 Age Structure

A serious concern about the results presented in this paper is that the variation in the WWII casualty rate originates in a difference in the age structure of the municipality population. Large cohorts of young men led to more drafts and more casualties relative to the total male population, even if the same share of drafted men did not return from the war. Of course the demography of a municipality population can have an effect on the post-war economic development, which would cause a serious omitted variable bias.

The preferred method to rule out an effect of the age structure of the population on economic development would be to control for the pre-war demographic composition of the municipality population or check the robustness of the results if the WWII casualty rate would be defined over the drafted men only. But the census of 1934 does not contain any age specific data to be used. Only in the census of 1939 and later some variables on the relative size of age groups are available. Finally, in the census of 1971 the population is divided into gender-specific 5-year age groups. As mentioned before, the census of 1939 does not account drafted soldiers to their home municipalities but to the municipality of their assigned barrack. The post-WWII censuses have the problem that they were, of course, affected by WWII casualties.

Notwithstanding these drawbacks, I use the age group variables to check the robustness of the main results when the share of certain age groups of the total municipality population are included in the regressions. If the high WWII casualty rate is only due to a large cohort of drafted men, then the estimated effect of the WWII casualties should shrink significantly if the size of the remaining age group after WWII is included in the specification. Table 11 reproduces the effects of WWII casualties on the size of the agricultural sector in 1951 (Panel A), the total wage bill in 2011 (Panel B), and mean wages in 2011 (Panel C) in column (I) for reference. In column (II) and (III) I include the share of the population of age 0 to 14 years, 14 to 18 years, and 18 to 65 years in 1951 and 1961, respectively. Column (IV) includes a set of control variables measuring the relative size of 5-year age groups of the male population in 1971.

In none of the specifications the estimated effect deviates much from the reference estimate in column (I), which suggests that the size of various cohorts either do not matter for economic development or it is not the driving force of the WWII casualty rate. In either case, it seems that the pre-war age structure of a municipality does not drive the results of the paper.

7 Alternative Mechanisms

I now discuss some potential mechanisms through which the WWII casualty rate could alternatively drive the results of the paper. However, the evidence presented here shows that these mechanisms were of second order and do not change the main results much. The potential mechanisms concern the education, fertility, and migration.

7.1 Education

Gennaioli *et al.* (2013, 2014) show that differences in human capital are the main explanatory factors of differences in income levels between regions of a country. The effect of human capital greatly outweighs differences in geography, institutions, and culture in these papers. It is therefore natural to inquire if the persistent effect of WWII casualties operates through a human capital channel. While Austrian municipalities feature minimal differences in geography, institutions, and culture, especially in within-district estimates, differences in educational attainment in the 1950s and 1960s could have set in motion a process of regional divergence.

A possible channel could be that children whose fathers did not return home after the war could not get higher education because of financial constraints. The consequently better educated population of low-casualty municipalities was able to establish firms and make better use of economic opportunities. A difference of the average education level could therefore be the link between WWII casualties and the level of economic activity today.

The highest level of education was first recorded in the population census of 1971. I use

these data to construct variables measuring the educational attainment and regress these on the WWII casualty rate. Table B.3 features the mean years of schooling as the dependent variable in Panel A, while Panel B uses the share of the population which only finished mandatory education. In none of these regressions there is a significant effect of WWII casualties on the educational outcome variable. All point estimates are small compared to the standard errors. Table B.4 shows that there is also no significant difference in the educational attainment in the age group of 20 to 34 years measured in 1971. That age group (birth cohorts between 1937 and 1951) have completed almost their entire education after WWII and should therefore give a clear picture if the educational attainment differed in high- vs. low-casualty municipalities.

To conclude, a difference in formal education does not seem to be the mechanism behind the long-run results of this paper. At least since the 1970s, there is no significant effect of WWII casualties on the education level of the residential population.

7.2 Fertility and Migration

The interruption in the gender-specific age distribution due to WWII casualties might also had an effect on the fertility of the local population that might influenced the local economy in the long-run. The birth cohorts with the highest casualty rates for all of Nazi-Germany were 1914–1924 (Overmans, 1999, p.234). These cohorts would have been between 21 and 29 years old at the end of the war and would likely have started a family in the immediate post-war period.

I explore this line of argument by examining the relationship between WWII casualties and population change (net birth and net migration) in the period 1961–1971 and the share of children in the population in the three population censuses following WWII. Table B.5 in the Appendix shows that the WWII casualty rate did not have a significant effect on the total population change between 1961 and 1971 (column I), net births (column II), or net migration (column III). The share of children (age 0–14 years) in 1951, 1961, and 1971 was also not significantly affected by war casualties as columns (IV)-(VI) show.

The results suggest that the local fertility pattern was not influenced by war casualties. The young population in neighboring municipalities seem to have had enough social interaction so that a local shock to a certain age-group of males could be absorbed.

Another line of argument for the estimated results could be that the population losses due to WWII in a municipality were absorbed by foreign refugees. If foreigners were less successful in generating economic activity, the long-term results found in this paper could also stem from this channel. This is especially relevant as there was a large inflow of refugees from the East into Austria in the years following WWII.

In Table B.6, I regress the share of foreigners of the total population in 1951, 1971, and 2001 on the WWII casualty rate, including the usual control variables and district fixed effects. There is no evidence of a concentration of foreigners in high-casualty municipalities.

8 Conclusion

This paper exploits military WWII casualties as a plausibly exogenous shock to local labor markets. In the short-run, high-casualty municipalities experience a shift of labor from agriculture towards manufacturing. Over the next 40 years, this initial difference in the sector labor allocation transforms into a lower share of workers in the service sector through the process of structural change. In the long-run, economic activity, measured by the total wage bill, is concentrated in municipalities with relatively few WWII casualties. This effect is large as a one percentage point increase in WWII casualties reduces the total wage bill in 2011 by almost 5 percent. The output reduction originates in, both, lower local wages and fewer workers.

The main contribution of the paper is the identification of a differential path of structural change as a, to the best of my knowledge, new mechanism through which past events can shape today's economic landscape. A temporary shock that alters the initial sector composition of the labor market can cause large long-term differences in the spatial distribution of economic activity. This, however, also implies that the temporary shock has to happen during a phase of structural change.

An important limitation to the results in this paper is that the sample consists of small and rural municipalities. These municipalities are small geographic areas and often consist of only one thousand people. Future work needs to show if the identified channel of persistence is also important in cities or larger forms of aggregation.

References

- ABRAMITZKY, R., DELAVANDE, A. and VASCONCELOS, L. (2011). Marrying Up: The Role of Sex Ratio in Assortative Matching. *American Economic Journal: Applied Economics*, 3 (3), 124–157.
- ACEMOGLU, D. (2010). When Does Labor Scarcity Encourage Innovation? Journal of Political Economy, **118** (6), 1037–1078.
- ---, HASSAN, T. and ROBINSON, J. (2011). Social Structure and Development: A Legacy of the Holocaust in Russia. *Quarterly Journal of Economics*, **126** (2), 895–946.
- BAUER, H., PALECZNY, A. and SCHULMEISTER, A. (1977). Aufgaben der Gemeinden. Wien: Institut für Stadtforschung.
- BAUER, T. K., BRAUN, S. and KVASNICKA, M. (2013). The Economic Integration of Forced Migrants: Evidence for Post-War Germany. *Economic Journal*, **123**, 998–1024.
- BRAKMAN, S., GARRETSEN, H. and SCHRAMM, M. (2004). The Strategic Bombing of Cities in Germany World War II and its Impact on City Growth. *Journal of Economic Geography*, 4 (2), 201–218.
- BRAUN, S. and KVASNICKA, M. (2014). Immigration and structural change: Evidence from post-war Germany. *Journal of International Economics*, **93** (2), 253–269.
- BUKEY, E. B. (2000). *Hitler's Austria: Popular Sentiment in the Nazi Era*, 1938-1955. Chapel Hill, N.C.: The University of North Carolina Press.
- DAVIS, D. R. and WEINSTEIN, D. E. (2002). Bones, Bombs, and Break Points: The Geography of Economic Activity. *American Economic Review*, **92** (5), 169–1289.
- and (2008). A Search for Multiple Equilibria in Urban Industrial Structure. Journal of Regional Science, 48 (1), 29–65.
- DELL, M. (2010). The Persistent Effects of Peru's Mining Mita. *Econometrica*, **78** (6), 1863–1903.
- DESMET, K. and ROSSI-HANSBERG, E. (2009). Spatial Growth and Industry Age. *Journal* of Economic Theory, 144, 2477–2502.
- EDER, C. and HALLA, M. (2015). The Long-lasting Shadow of the Allied Occupation of Austria on its Spatial Equilibrium, unpublished manuscript.
- GENNAIOLI, N., LA PORTA, R., LOPEZ DE SILANES, F. and SHLEIFER, A. (2013). Human Capital and Regional Development. *Quarterly Journal of Economics*, **128** (1), 105–164.
- —, —, and SHLEIFER, A. (2014). Growth in Regions. Journal of Economic Growth, **19** (3), 259–309.

- GREENE, W. (2008). *Econometric Analysis*. Upper Saddle River, New Jersey: Pearson Prentice Hall, 6th edn.
- HABAKKUK, H. (1962). American and British Technology in the Nineteenth Century. London: Cambridge University Press.
- HAGSPIEL, H. (1995). Die Ostmark: Österreich im Grossdeutschen Reich 1938 bis 1945. Vienna: Braumüller.
- HERRENDORF, B., ROGERSON, R. and KOS VALENTINYI (2014). Growth and Structural Transformation. In P. Aghion and S. Durlauf (eds.), *Handbook of Economic Growth, Chap*ter 6, vol. 2B, North Holland: Elsevier.
- HORNBECK, R. and NAIDU, S. (2014). When the Levee Breaks: Black Migration and Economic Development in the American South. *American Economic Review*, **104** (3), 963–990.
- KRUGMAN, P. (1991). Increasing Returns and Economic Geography. Journal of Political Economy, 99 (3), 483–499.
- and VENABLES, A. (1995). Globalization and the Inequality of Nations. Quarterly Journal of Economics, 110 (4), 857–880.
- KUZNETS, S. (1957). Quantitative Aspects of the Economic Growth of Nations: II. Industrial Distribution of National Product and Labor Force. *Economic Development and Cultural Change*, 5 (4), 1–111.
- MATSUYAMA, K. (1992). A Simple Model of Sectoral Adjustment. *Review of Economic Studies*, **59**, 375–388.
- MIGUEL, E. and ROLAND, G. (2011). The Long-run Impact of Bombing Vietnam. *Journal* of Development Economics, 96 (1), 1–15.
- MORAWEK, E. and NEUGEBAUER, W. (1989). Osterreicher und der Zweite Weltkrieg. Wien: Österreichischer Bundesverlag.
- MORETTI, E. (2011). Local Labor Markets. In D. Card and O. Ashenfelter (eds.), *Handbook* of Labor Economics, vol. 4B, 14, Elsevier B.V., pp. 1237–1313.
- NUNN, N. (2014). Historical Development. In P. Aghion and S. Durlauf (eds.), Handbook of *Economic Growth, Chapter 7*, vol. 2A, North Holland: Elsevier.
- OVERMANS, R. (1999). Deutsche militärische Verluste im Zweiten Weltkrieg. München: R. Oldenbourg Verlag.
- RYBCZYNSKI, T. M. (1955). Factor Endowment and Relative Commodity Prices. *Econom*ica, **22** (88), 336–341.
- SARVIMÄKI, M., UUSITALO, R. and JÄNTTI, M. (2016). Habit Formation and the Misallocation of Labor: Evidence from Forced Migrations, unpublished manuscript.

- SCHUMANN, A. (2014). Persistence of Population Shocks: Evidence from the Occupation of Wets Germany after World War II. American Economic Journal: Applied Economics, 6 (3), 189–205.
- STANEK, E. (1985). Verfolgt, verjagt, vertrieben: Flüchtlinge in Österreich von 1945–1984. Europaverlag.
- THALMANN, F. (1954). Die Wirtschaft in Österreich. In H. Benedikt (ed.), Geschichte der Republik Österreich, Wien: Verlag für Geschichte und Politik.
- VOCELKA, K. (2010). Österreichische Geschichte. München: C.H.Beck.
- WALDINGER, F. (forthcoming). Bombs, Brains, and Science: The Role of Human and Physical Capital for the Creation of Scientific Knowledge. *Review of Economics and Statistics*.
- ZWEIMÜLLER, J., WINTER-EBMER, R., LALIVE, R., KUHN, A., WUELLRICH, J.-P., RUF, O. and BÜCHI, S. (2009). *Austrian Social Security Database*. Tech. rep., The Austrian Center for Labor Economics and the Analysis of the Welfare State.

Tables

Variable	Obs	Mean	Std. Dev.	Min	Max
PANEL A: WWIL cognettion					
FAINEL A. WWII casuallies					
% WWII casualties of 1934 male pop.	314	0.117	0.030	0.052	0.220
WWII casualties	314	69.318	59.521	6	483
Population in 1934	314	1,180.057	983.179	73	8,733
Population in 2011	314	2,200.873	2,401.980	53	21,181
PANEL B: Control variables					
Log population in 1934	314	6.770	0.802	4.29	9.075
% population in agriculture in 1934	314	0.571	0.248	0.008	0.971
Market status in 1945 (dummy)	314	0.178	0.383	0	1
Log elevation	314	6.531	0.840	4.761	7.762
Log slope	314	2.322	1.221	-1.122	3.583
$Log elevation \times log slope$	314	16.096	9.020	-5.347	27.172
% Jewish population in 1934	314	0.001	0.007	0	0.117
% Social Democrats in 1930	314	0.202	0.193	0	0.710
% NSDAP in 1930	314	0.013	0.027	0	0.247
PANEL C: Sector shares					
% workers in agriculture in 1951	314	0.520	0.241	0.012	0.974
% workers in agriculture in 1981	314	0.120	0.113	0.004	0.603
% workers in agriculture in 2011	314	0.052	0.049	0.004	0.360
% workers in manufacturing in 1951	314	0.328	0.179	0.026	0.869
% workers in manufacturing in 1981	314	0.413	0.131	0.074	0.730
% workers in manufacturing in 2011	314	0.268	0.076	0.045	0.468
% workers in services in 1951	314	0.152	0.093	0	0.518
% workers in services in 1981	314	0.467	0.130	0.155	0.881
% workers in services in 2011	314	0.680	0.086	0.390	0.898
PANEL D: Economic outcomes					
Log total wage bill in 2011	314	8.809	1.542	4.200	12.140
Log mean yearly wage in 2011	312	10.051	0.239	9.228	10.580
Log local workers in 2011	314	5.961	1.229	2.079	9.191
Log local workers in agriculture in 2011	314	3.404	0.918	0	5.914
Log local workers in manufact. in 2011	314	4.526	1.620	0	8.034
Log local workers in services in 2011	314	5.556	1.302	1.609	8.892
Log number of firms in 2011	314	4.821	0.900	1.609	7.330
Log number of firms in agricult. in 2011	314	2.976	0.867	0	5.394
Log number of firms in manufact. in 2011	314	2.715	1.054	0	5.733
Log number of firms in services in 2011	314	4.360	1.049	0.693	7.061

 Table 1:
 Summary Statistics of Main Variables

I Casualtie
IWV
5
and
stics
aracteri
Ch
ality
icip
Mun
Var]
Pre-V
с:

Table 2: Pre	-War Mı	unicipalit	y Charae	cteristics	and WW	/II Casual	lties	
				WWII ce	asualty rate	Ð		
	(I)	(II)	(III)	(IV)	(V)	(VI)	(III)	(IIII)
Log population in 1934	-0.004						-0.003	-0.002
	(0.003)						(0.004)	(0.004)
% pop. in agriculture in 1934		0.013					0.016	0.013
		(0.008)	100 0				(0.014)	(0.014)
Market status in 1945 (dummy)			(0.002)				-0.004 (0.006)	-0.002
Log elevation				0.003			0.006	0.006
				(0.018)			(0.018)	(0.018)
Log slope				0.014			0.021	0.020
				(0.021)			(0.022)	(0.022)
Log elevation \times log slope				-0.002			-0.004	-0.004
				(0.004)			(0.004)	(0.004)
% Jewish pop. in 1934					-0.458^{*}		-0.353	-0.372
					(0.237)		(0.243)	(0.242)
% Social Democrats in 1930						-0.016	0.005	0.004
						(0.013)	(0.018)	(0.018)
% NSDAP in 1930						0.132	0.186^{**}	0.064
						(0.089)	(0.094)	(0.121)
District FE	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	${\rm Yes}$	\mathbf{Yes}
NSDAP outlier excluded								\mathbf{Yes}
No. observations	314	314	314	314	314	314	314	313
R-squared	0.24	0.24	0.23	0.23	0.24	0.24	0.27	0.25
Mean of dep. var.	0.12	0.12	0.12	0.12	0.12	0.12	0.12	0.12
S.d. of dep. var.	0.03	0.03	0.03	0.03	0.03	0.03	0.03	0.03
P-value	0.104	0.109	0.328	0.824	0.054	0.180	0.197	0.503
This table summarizes estimation	results ba	um no bea	nicipality-	level data.	The depe	ndent varia	ble is equal t	o military
WWII casualties divided by the r	male muni	cipality po	pulation i	n 1934. Tl	he control	variables in	each specific	cation are
listed. In addition, district fixed	effects ar	e included	. The me	thod of es	timation is	s least squa	res. Robust	standard
errors are in parentheses below. I	he row "F	-value" re	ports the] * 202 ***	p-values of	a two-side	d hypothes	is test of the	estimates
or all listed control variables bell level, and 1% level, respectively.	ug jounuy	zero. ', '		mancate	statistical	signincance	at the 10%	level, 370

		r	WWII casi	ualty rate		
	(I)	(II)	(III)	(IV)	(V)	(VI)
Independent variable: % of age group 55–64 in 1971 with						
only mandatory education	0.019 (0.020)	0.001 (0.026)				
apprenticeship	. ,	. ,	-0.011 (0.030)	0.020 (0.038)		
secondary schooling			(0.000)	(0.000)	-0.055 (0.041)	-0.030 (0.046)
Control variables		Yes		Yes		Yes
District FE	Yes	Yes	Yes	Yes	Yes	Yes
No. observations	314	314	314	314	314	314
R-squared	0.23	0.27	0.23	0.27	0.24	0.27
Mean of dep. var.	0.12	0.12	0.12	0.12	0.12	0.12
S.d. of dep. var.	0.03	0.03	0.03	0.03	0.03	0.03

Table 3: Pre-War Education and WWII Casualties

This table summarizes estimation results based on municipality-level data. The dependent variable is equal to military WWII casualties divided by the male municipality population in 1934. The education variables of the age group 55–64 is measured in 1971. The control variables (if included) in each specification are listed in Panel B of Table 1. In addition, district fixed effects are included. The method of estimation is least squares. Robust standard errors are in parentheses below. *, ** and *** indicate statistical significance at the 10% level, 5% level, and 1% level, respectively.

	Log Population in 1900 (I)	Log Horses in 1900 (II)	Log Cattle in 1900 (III)	Log Pigs in 1900 (IV)	Log Sheep in 1900 (V)	Log Large Land holdings in 1900 (VI)	Log Factories in 1900 (VII)
WWII casualty rate	0.536 (0.359)	-0.486 (1.861)	-0.691 (0.727)	0.390 (1.201)	4.159 (3.182)	0.269 (1.211)	-0.186 (0.686)
Controls District FE	${ m Yes}{ m Yes}$	${ m Yes}_{ m Yes}$	${ m Yes}_{ m Yes}$	${ m Yes}{ m Yes}$	${ m Yes}_{ m es}$	Yes Yes	${ m Yes}{ m Yes}$
No. observations R_sourced	314 0 06	$\begin{array}{c} 259\\ 0 \ 76 \end{array}$	$\begin{array}{c} 259\\ 0.81 \end{array}$	259 0 79	$\begin{array}{c} 259 \\ 0 \ 71 \end{array}$	$\begin{array}{c} 259\\ 0.53\end{array}$	259 0.67
Mean of dep. var.	6.62	2.87	6.16	4.64	3.30	0.41	0.18
S.d. of dep. var. P-value	$\begin{array}{c} 0.78 \\ 0.137 \end{array}$	$\begin{array}{c} 1.45\\ 0.794\end{array}$	$0.64 \\ 0.343$	$\begin{array}{c} 0.99 \\ 0.746 \end{array}$	$2.25 \\ 0.193$	0.67 0.825	$\begin{array}{c} 0.45\\ 0.787\end{array}$
This table summaric column heading. Theffects are included. row "P-value" repor * ** and *** indica	tes estimation result te control variable The method of e ts the p-values of te statistical signifi	ilts based on s in each spec stimation is l a two-sided h	municipality cification are east squares. ypothesis tea	-level data. • listed in P • Robust st st of the est % level and	The depend anel B of Ta andard error imate of the	ent variable is mer ble 1. In addition, s are in parenthese WWII casualty ra	ationed in the district fixed se below. The te being zero.

Casualties
IIWW
ment and
Develop
Pre-War
Table 4:

	Share of	Diff. in log	Share of family-	Share of family-
	workers in	pop. in agric.	internal agric.	external agric.
	agriculture	btw. 1934	workers of pop.	workers of pop.
	in 1951	and 1951	in 1960	in 1960
	(I)	(II)	(III)	(IV)
WWII casualty rate	-0.490^{***}	-1.050^{***}	-0.233^{*}	-0.089^{**}
	(0.138)	(0.376)	(0.121)	(0.044)
Control variables	Yes	Yes	Yes	Yes
District FE	Yes	Yes	Yes	Yes
No. observations	314	314	314	314
R-squared	0.94	0.33	0.84	0.47
Mean of dep. var.	0.52	-0.18	0.20	0.02
S.d. of dep. var.	0.24	0.19	0.13	0.03

Table 5: Short-Run Effects of WWII Casualties on Agricultural Outcomes

This table summarizes estimation results based on municipality-level data. The dependent variable is mentioned in the column heading. The control variables in each specification are listed in Panel B of Table 1. In addition, district fixed effects are included. The method of estimation is least squares. Robust standard errors are in parentheses below. *, ** and *** indicate statistical significance at the 10% level, 5% level, and 1% level, respectively.

	Table 6:	WWII Casu	alties and the	Evolution of	f Sector Share	SS	
PANEL A	1951	1961	Share of we 1971	orkers in agricul 1981	lture in 1991	2001	2011
WWII casualty rate	-0.490^{***} (0.138)	-0.289 (0.180)	-0.358* (0.204)	$0.040 \\ (0.149)$	-0.016 (0.107)	0.111 (0.090)	0.049 (0.071)
R-squared Mean of dep. var. S.d. of dep. var.	$\begin{array}{c} 0.94 \\ 0.52 \\ 0.24 \end{array}$	$\begin{array}{c} 0.89\\ 0.39\\ 0.23\end{array}$	$\begin{array}{c} 0.76 \\ 0.21 \\ 0.17 \end{array}$	$\begin{array}{c} 0.70 \\ 0.12 \\ 0.11 \end{array}$	0.68 0.08 0.08	0.61 0.06 0.06	0.65 0.05 0.05
PANEL B	1951	1961	Share of worl 1971	kers in manufac 1981	turing in 1991	2001	2011
WWII casualty rate	0.573^{***}	0.592^{***}	0.656^{***}	0.517^{***}	0.459^{***}	0.422*** (0.150)	0.249^{**}
R-squared Mean of dep. var. S.d. of dep. var.	0.33 0.18 0.18	0.79 0.37 0.18	0.66	0.64 0.41 0.13	0.58 0.37 0.11	0.52 0.30 0.09	0.54 0.27 0.08
PANEL C	1951	1961	Share of v 1971	workers in servi 1981	ces in 1991	2001	2011
WWII casualty rate R-squared Mean of dep. var. S.d. of dep. var.	$\begin{array}{c} -0.084 \\ (0.126) \\ 0.69 \\ 0.15 \\ 0.09 \end{array}$	$\begin{array}{c} -0.303 \\ (0.190) \\ 0.60 \\ 0.24 \\ 0.12 \end{array}$	$\begin{array}{c} -0.299 \\ (0.225) \\ 0.57 \\ 0.36 \\ 0.14 \end{array}$	-0.557^{***} (0.199) 0.60 0.47 0.13	-0.443^{**} (0.183) 0.54 0.55 0.11	-0.532^{***} (0.165) 0.51 0.64 0.10	$\begin{array}{c} -0.298^{**} \\ (0.130) \\ 0.61 \\ 0.68 \\ 0.09 \end{array}$
Control variables District FE No. observations	$\begin{array}{c} {\rm Yes} \\ {\rm Yes} \\ 314 \end{array}$	$\begin{array}{c} \mathrm{Yes} \\ \mathrm{Yes} \\ 314 \end{array}$	$\begin{array}{c} \mathrm{Yes} \\ \mathrm{Yes} \\ 314 \end{array}$	$\begin{array}{c} \mathrm{Yes} \\ \mathrm{Yes} \\ 314 \end{array}$	$\begin{array}{c} \mathrm{Yes} \\ \mathrm{Yes} \\ 314 \end{array}$	Yes Yes 314	Yes Yes 314
This table summarizes heading. The control included. The method statistical significance	s estimation res variables in es l of estimation at the 10% lev	ults based on m ach specificatior is least squares. el, 5% level, and	unicipality-leve 1 are listed in I Robust stands 1 1% level, resp.	l data. The der Panel B of Tab ard errors are ir ectively.	pendent variable de 1. In additi 1 parentheses be	; is mentioned in on, district fixe elow. *, ** and	a the column ed effects are *** indicate

		Log t	otal wage bill in	n 2011	
	(I)	(II)	(III)	(IV)	(V)
WWII casualty rate	-6.901^{**}	-4.825^{**}	-5.038^{**}	-4.772^{***}	-5.671^{**}
T 1.4. 1094	(2.880)	(2.013)	(1.964)	(1.689)	(2.193)
Log population in 1934		(0.103)	(0.114)		(0.114)
Log population in 2011		~ /		1.237^{***}	
WWII casualty rate \times Soviet zone				(0.003)	3.135
% pop. in a griculture in 1934		-2.822***	-2.453^{***}	-1.684^{***}	(4.816) -2.467^{***}
Market status in 1045 (dummy)		(0.375) 0.271	(0.442)	(0.386)	(0.443) 0.077
Market status in 1945 (duniny)		(0.183)	(0.193)	(0.165)	(0.193)
Log elevation		0.662^{**}	0.546	0.651	0.534
Log slope		(0.320) -0.115	0.678	0.557	0.647
Log elevation \times log slope		(0.473) -0.039	(0.691) -0.174	$(0.594) \\ -0.123$	(0.693) -0.169
% Jewish pop. in 1934		$(0.085) \\ -0.983$	(0.126) -2.261	$\begin{array}{c}(0.108)\\0.953\end{array}$	(0.126) -1.258
% Social Democrats in 1030		(8.563) -2 147***	(7.709) -1 653***	(6.638) -1 298***	(7.870) -1 665***
70 Social Democrats in 1950		(0.474)	(0.571)	(0.492)	(0.572)
% NSDAP in 1930		$\begin{array}{c} 0.581 \ (2.351) \end{array}$	$1.162 \\ (3.001)$	0.481 (2.579)	$1.163 \\ (3.004)$
District FE			Yes	Yes	Yes
No. observations	314	314	314	314	314
R-squared	0.02	0.58	0.72	0.80	0.72
Mean of dep. var.	8.81	8.81	8.81	8.81	8.81
S.d. of dep. var.	1.54	1.54	1.54	1.54	1.54

Table 7: WWII Casualties and Long-Run Development I

This table summarizes estimation results based on municipality-level data. The dependent variable is mentioned in the column heading. The control variables in each specification are listed. In addition, district fixed effects are included. The method of estimation is least squares. Robust standard errors are in parentheses below. *, ** and *** indicate statistical significance at the 10% level, 5% level, and 1% level, respectively.

PANEL A		Log	mean yearly v	vages	
	1972	1981	1991	2001	2011
	(I)	(II)	(III)	(IV)	(V)
WWII casualty rate	-0.772^{*}	-1.144^{***}	-1.512^{***}	-1.513^{***}	-1.396^{***}
	(0.424)	(0.338)	(0.334)	(0.446)	(0.421)
Control variables	Yes	Yes	Yes	Yes	Yes
District FE	Yes	Yes	Yes	Yes	Yes
No. observations	275	300	311	313	312
R-squared	0.59	0.61	0.59	0.54	0.63
Mean of dep. var.	8.45	9.27	9.70	9.97	10.19
S.d. of dep. var.	0.18	0.16	0.15	0.20	0.21
PANEL B		Log mea	an yearly wage	s in 2011	
	Males	Females	Agriculture	Manufacturing	Services
	(I)	(II)	(III)	(IV)	(V)
WWIII equalty note	1 966***	1 900***	1.054	1 100**	1 005***
w will casualty rate	-1.300	-1.290	(1.054)	-1.180^{-1}	-1.083
	(0.408)	(0.380)	(1.507)	(0.484)	(0.387)
Control variables	Yes	Yes	Yes	Yes	Yes
District FE	Yes	Yes	Yes	Yes	Yes
No. observations	311	312	187	298	312
R-squared	0.67	0.58	0.67	0.52	0.54
Mean of dep. var.	10.36	9.90	9.06	10.37	10.08
S.d. of dep. var.	0.21	0.18	0.52	0.19	0.17

Table 8: WWII Casualties and Long-Run Development II

This table summarizes estimation results based on municipality-level data. The dependent variable is mentioned in the column heading. The control variables in each specification are listed in Panel B of Table 1. In addition, district fixed effects are included. Each observation is weighted by the number of workers that mean wage is based on. The method of estimation is least squares. Robust standard errors are in parentheses below. *, ** and *** indicate statistical significance at the 10% level, 5% level, and 1% level, respectively.

PANEL A	Log local workers					
	Total	Agriculture	Manufacturing	Services		
	(I)	(II)	(III)	(IV)		
WWII casualty rate	-2.827^{**}	0.059	-0.552	-3.308^{**}		
	(1.268)	(1.161)	(2.119)	(1.364)		
Control variables	Yes	Yes	Yes	Yes		
District FE	Yes	Yes	Yes	Yes		
No. observations	314	314	314	314		
R-squared	0.82	0.73	0.71	0.81		
Mean of dep. var.	5.96	3.40	4.53	5.56		
S.d. of dep. var.	1.23	0.92	1.62	1.30		
PANEL B	-	Log nur	mber of firms	~ .		
	Total	Agriculture	Manufacturing	Services		
	(I)	(II)	(III)	(IV)		
WWII casualty rate	-1.887**	0.297	-1.366	-2.675**		
WWII casualty rate	-1.887^{**} (0.845)	$0.297 \\ (1.015)$	-1.366 (1.075)	-2.675^{**} (1.056)		
WWII casualty rate Control variables	-1.887** (0.845) Yes	0.297 (1.015) Yes	-1.366 (1.075) Yes	-2.675^{**} (1.056) Yes		
WWII casualty rate Control variables District FE	-1.887** (0.845) Yes Yes	0.297 (1.015) Yes Yes	-1.366 (1.075) Yes Yes	-2.675** (1.056) Yes Yes		
WWII casualty rate Control variables District FE No. observations	-1.887** (0.845) Yes Yes 314	0.297 (1.015) Yes Yes 314	-1.366 (1.075) Yes Yes 314	-2.675^{**} (1.056) Yes Yes 314		
WWII casualty rate Control variables District FE No. observations R-squared	-1.887^{**} (0.845) Yes Yes 314 0.85	0.297 (1.015) Yes Yes 314 0.77	-1.366 (1.075) Yes Yes 314 0.82	-2.675^{**} (1.056) Yes Yes 314 0.83		
WWII casualty rate Control variables District FE No. observations R-squared Mean of dep. var.	$-1.887^{**} \\ (0.845)$ Yes Yes $314 \\ 0.85 \\ 4.82$	$\begin{array}{c} 0.297 \\ (1.015) \end{array}$ Yes Yes $314 \\ 0.77 \\ 2.98 \end{array}$	-1.366 (1.075) Yes Yes 314 0.82 2.72	$\begin{array}{c} -2.675^{**} \\ (1.056) \end{array}$ Yes Yes 314 0.83 4.36		

Table 9: WWII Casualties and Long-Run Development III

This table summarizes estimation results based on municipality-level data. The dependent variable is mentioned in the column heading. The control variables in each specification are listed in Panel B of Table 1. In addition, district fixed effects are included. The method of estimation is least squares. Robust standard errors are in parentheses below. *, ** and *** indicate statistical significance at the 10% level, 5% level, and 1% level, respectively.

		Log total wage	bill in 2011	
	(I)	(II)	(III)	(IV)
WWII casualty rate	-5.355^{***}	-5.021^{***}	-5.279^{***}	-5.858^{***}
	(1.716)	(1.742)	(1.839)	(1.918)
Control variables	Yes	Yes	Yes	Yes
Pair FE	Yes	Yes	Yes	Yes
Sample I		Yes		
Sample II			Yes	
Sample III				Yes
No. observations	652	638	564	526
R-squared	0.88	0.88	0.88	0.86
Mean of dep. var.	8.99	9.00	8.97	8.85
S.d. of dep. var.	1.55	1.55	1.55	1.47

Table 10: Robustness Check: WWII Casualties and Local Economic Activity

This table summarizes estimation results based on municipality-level data. The dependent variable is mentioned in the column heading. The control variables in each specification are listed in Panel B of Table 1. In addition, pair fixed effects (where pairs are given by neighboring municipalities) are included. The method of estimation is least squares. Robust standard errors are in parentheses below. *, ** and *** indicate statistical significance at the 10% level, 5% level, and 1% level, respectively.

PANEL A	Share of workers in agriculture in 1951					
	(I)	(II)	(III)	(IV)	(V)	
WWII casualty rate	-0.49^{***} (0.14)	-0.43^{***} (0.14)	-0.48^{***} (0.14)	-0.49^{***} (0.14)	-0.49^{***} (0.14)	
R-squared Mean of dep. var. S.d. of dep. var.	$0.94 \\ 0.52 \\ 0.24$	$0.95 \\ 0.52 \\ 0.24$	$0.95 \\ 0.52 \\ 0.24$	$0.95 \\ 0.52 \\ 0.24$	$0.95 \\ 0.52 \\ 0.24$	
PANEL B		Log tota	l wage bill in	2011		
	(I)	(II)	(III)	(IV)	(V)	
WWII casualty rate	-5.04^{**} (1.96)	-5.03^{**} (1.96)	-5.34^{***} (1.93)	-5.35^{***} (1.96)	-4.58^{**} (1.92)	
R-squared Mean of dep. var. S.d. of dep. var.	$0.72 \\ 8.81 \\ 1.54$	$0.73 \\ 8.81 \\ 1.54$	$0.74 \\ 8.81 \\ 1.54$	$0.73 \\ 8.81 \\ 1.54$	$0.77 \\ 8.81 \\ 1.54$	
PANEL C	Log mean yearly wages in 2011					
	(I)	(II)	(III)	(IV)	(V)	
WWII casualty rate	-1.40^{***} (0.42)	-1.43^{***} (0.42)	-1.32^{***} (0.42)	-1.38^{***} (0.42)	-1.38^{***} (0.42)	
R-squared Mean of dep. var. S.d. of dep. var.	$0.63 \\ 10.19 \\ 0.21$	$0.64 \\ 10.19 \\ 0.21$	$0.64 \\ 10.19 \\ 0.21$	$0.63 \\ 10.19 \\ 0.21$	$0.68 \\ 10.19 \\ 0.21$	
Age structure 1939 controls Age structure 1951 controls Age structure 1961 controls Age structure 1971 controls		Yes	Yes	Yes	Veg	
Controls District FE No. observations	Yes Yes 314	Yes Yes 314	Yes Yes 314	Yes Yes 314	Yes Yes 314	

 Table 11:
 Age Structure Control Variables

This table summarizes estimation results based on municipality-level data. The dependent variable is mentioned in the column heading. Different control variables for the age structure in 1939, 1951, 1961 or 1971 ar included in columns (II)-(V), respectively. The usual control variables in each specification are listed in Panel B of Table 1. The method of estimation is least squares. Robust standard errors are in parentheses below. *, ** and *** indicate statistical significance at the 10% level, 5% level, and 1% level, respectively.

For Online Publication

Appendix A Additional Figures

Figure A.1: Male and Female Population by Birth Cohort in Austria



The horizontal axis shows the year of birth and the vertical axis the population in thousands. The title of each diagram reports the year of the census. The grey bars mark WWI and WWII. Data is from the Austrian population censuses.



Figure A.2: Correlations of the WWII Casualty Rate with Pre-war Municipality Characteristics

Appendix B Additional Tables

Log yearly wages in 2011						
	(I)	(II)	(III)	(IV)	(V)	
WWII casualty rate	-1.919^{***} (0.675)	-1.725^{***} (0.485)	-1.328^{***} (0.380)	-1.748^{***} (0.664)	-1.141^{***} (0.316)	
Municipality controls	Yes	Yes	Yes	Yes	Yes	
Individual controls		Yes			Yes	
Firm controls			Yes		Yes	
Job controls				Yes	Yes	
District FE	Yes	Yes	Yes	Yes	Yes	
No. observations	146,540	146,540	146,540	146,540	146,540	
R-squared	0.07	0.22	0.24	0.11	0.36	
Mean of dep. var.	9.99	9.99	9.99	9.99	9.99	
S.d. of dep. var.	0.79	0.79	0.79	0.79	0.79	

Table B.1: WWII Casualties and Individual Wages

This table summarizes estimation results based on individual-level data. The dependent variable is mentioned in the column heading. The set of control variables in each specification are indicated. Municipality control variables are listed in Panel B of Table 1. Individual control variables are dummy variables for each year of age, gender, highest education level achieved, and foreigners. Firm controls are industry indicators at the two-digit NACE level. Job controls are indicators for apprentices, blue and white collar jobs. In addition, district fixed effects are included. The method of estimation is least squares. Standard errors are clustered at the municipality level and are shown in parentheses below. *, ** and *** indicate statistical significance at the 10% level, 5% level, and 1% level, respectively.

	Log mean yearly wages in 2011					
	(I)	(II)	(III)	(IV)		
WWII casualty rate	-1.332^{***}	-1.327^{***}	-1.413^{***}	-0.833**		
·	(0.335)	(0.342)	(0.366)	(0.391)		
Control variables	Yes	Yes	Yes	Yes		
Pair FE	Yes	Yes	Yes	Yes		
Sample I		Yes				
Sample II			Yes			
Sample III				Yes		
No. observations	649	635	561	524		
R-squared	0.91	0.90	0.91	0.90		
Mean of dep. var.	10.21	10.21	10.21	10.17		
S.d. of dep. var.	0.21	0.21	0.22	0.22		

Table B.2: Robustness Check: WWII Casualties and Wages

This table summarizes estimation results based on municipality-level data. The dependent variable is mentioned in the column heading. The control variables in each specification are listed in Panel B of Table 1. In addition, pair fixed effects (where pairs are given by neighboring municipalities) are included. Each observation is weighted by the number of workers that mean wage is based on. The method of estimation is least squares. Robust standard errors are in parentheses below. *, ** and *** indicate statistical significance at the 10% level, 5% level, and 1% level, respectively.

PANEL A	Mean years of education of the pop. 15+					
	1971	1981	1991	2001	2011	
	(I)	(II)	(III)	(IV)	(V)	
WWII casualty rate	0.298	0.379	0.413	-0.312	-0.183	
	(0.593)	(0.628)	(0.628)	(0.610)	(0.627)	
Control variables	Yes	Yes	Yes	Yes	Yes	
District FE	Yes	Yes	Yes	Yes	Yes	
No. observations	314	314	314	314	314	
R-squared	0.67	0.64	0.60	0.59	0.57	
Mean of dep. var.	10.04	10.64	11.21	11.69	12.16	
S.d. of dep. var.	0.43	0.43	0.41	0.39	0.40	
PANEL B	Share of pop. 15+ with only mandatory education					
	1971	1981	1991	2001	2011	
	(I)	(II)	(III)	(IV)	(V)	
WWII casualty rate	-0.066	-0.098	-0.073	0.061	0.037	
	(0.141)	(0.139)	(0.130)	(0.105)	(0.089)	
Control variables	Yes	Yes	Yes	Yes	Yes	
District FE	Yes	Yes	Yes	Yes	Yes	
No. observations	314	314	314	314	314	
R-squared	0.67	0.63	0.57	0.54	0.51	
Mean of dep. var.	0.75	0.60	0.47	0.38	0.30	
S.d. of dep. var.	0.10	0.09	0.08	0.06	0.05	

Table B.3: WWII Casualties and Educational Outcomes

This table summarizes estimation results based on municipality-level data. The dependent variable is mentioned in the column heading. The control variables in each specification are listed in Panel B of Table 1. In addition, district fixed effects are included. The method of estimation is least squares. Robust standard errors are in parentheses below. *, ** and *** indicate statistical significance at the 10% level, 5% level, and 1% level, respectively.

	Share of 20 only mandatory education	to 34 year olds apprentice- ship	in 1971 with secondary schooling
	(1)	(11)	(111)
WWII casualty rate	-0.285	0 252	0.016
W WII cabacity face	(0.211)	(0.159)	(0.120)
Control variables	Yes	Yes	Yes
District FE	Yes	Yes	Yes
No. observations	314	314	314
R-squared	0.61	0.64	0.35
Mean of dep. var.	0.59	0.27	0.12
S.d. of dep. var.	0.14	0.11	0.06

Table B.4: WWII Casualties and Age-Specific Educational Outcomes in 1971

This table summarizes estimation results based on municipality-level data. The dependent variable is mentioned in the column heading. The control variables in each specification are listed in Panel B of Table 1. In addition, district fixed effects are included. The method of estimation is least squares. Robust standard errors are in parentheses below. *, ** and *** indicate statistical significance at the 10% level, 5% level, and 1% level, respectively.

	Change in population 1961-71 Total Natural Migration			Share of pop. 0–14 years old in 1051 1061 1071		
	(I)	(II)	(III)	(IV)	(V)	(VI)
WWII casualty rate	-0.011	0.136	-0.147	-0.003	0.077	-0.044
	(0.253)	(0.115)	(0.210)	(0.065)	(0.066)	(0.063)
Control variables	Yes	Yes	Yes	Yes	Yes	Yes
District FE	Yes	Yes	Yes	Yes	Yes	Yes
No. observations	314	314	314	314	314	314
R-squared	0.58	0.68	0.52	0.58	0.62	0.72
Mean of dep. var.	0.14	0.13	0.01	0.25	0.27	0.28
S.d. of dep. var.	0.16	0.08	0.13	0.04	0.04	0.05

Table B.5:Population Changes

This table summarizes estimation results based on municipality-level data. The dependent variable is mentioned in the column heading. The control variables in each specification are listed in Panel B of Table 1. In addition, district fixed effects are included. The method of estimation is least squares. Robust standard errors are in parentheses below. *, ** and *** indicate statistical significance at the 10% level, 5% level, and 1% level, respectively.

	Share of foreigners in population				
	1951	1971	2001		
	(I)	(II)	(III)		
WWII casualty rate	-0.001	-0.055	-0.015		
	(0.048)	(0.047)	(0.077)		
Control variables	Yes	Yes	Yes		
District FE	Yes	Yes	Yes		
No. observations	314	314	314		
R-squared	0.43	0.56	0.56		
Mean of dep. var.	0.03	0.02	0.07		
S.d. of dep. var.	0.03	0.03	0.05		

Table B.6: Foreigner population

This table summarizes estimation results based on municipality-level data. The dependent variable is mentioned in the column heading. The control variables in each specification are listed in Panel B of Table 1. In addition, district fixed effects are included. The method of estimation is least squares. Robust standard errors are in parentheses below. *, ** and *** indicate statistical significance at the 10% level, 5% level, and 1% level, respectively.

Appendix C Data Sources

The data for the economic variables come from various publications of *Statistik Austria*, the Austrian statistical agency (earlier names of the agency: *K. K. Statistischen Zentralkommission, Bundesamt für Statistik*, and Österreichisches Statistisches Zentralamt):

- "Gemeindelexikon der im Reichsrate vertretenen Königreiche und Länder: Bearbeitet auf Grund der Ergebnisse der Volkszählung vom 31. Dezember 1900"
- "Die Ergebnisse der österreichischen Volkszählung vom 22. März 1934"
- "Die Ergebnisse der österreichischen Volkszählung vom 1. Juni 1951"
- "Ergebnisse der land- und forstwirtschaftlichen Betriebszählung vom 1. Juni 1951"
- "Die Ergebnisse der österreichischen Volkszählung vom 21. März 1961"
- "Land- und forstwirtschaftliche Betriebszählung vom 1. Juni 1960"
- "Die Ergebnisse der österreichischen Volkszählung vom 12. Mai 1971"
- "Volkszählung 1981"
- "Volkszählung 1991"
- "Ein Blick auf die Gemeinde" (www.statistik.at/blickgem)

University of Innsbruck - Working Papers in Economics and Statistics Recent Papers can be accessed on the following webpage:

http://eeecon.uibk.ac.at/wopec/

- 2016-22 Christoph Eder: Missing men: World War II casualties and structural change
- 2016-21 Reto Stauffer, Jakob Messner, Georg J. Mayr, Nikolaus Umlauf, Achim Zeileis: Ensemble post-processing of daily precipitation sums over complex terrain using censored high-resolution standardized anomalies
- 2016-20 Christina Bannier, Eberhard Feess, Natalie Packham, Markus Walzl: Incentive schemes, private information and the double-edged role of competition for agents
- 2016-19 Martin Geiger, Richard Hule: Correlation and coordination risk
- 2016-18 Yola Engler, Rudolf Kerschbamer, Lionel Page: Why did he do that? Using counterfactuals to study the effect of intentions in extensive form games
- 2016-17 Yola Engler, Rudolf Kerschbamer, Lionel Page: Guilt-averse or reciprocal? Looking at behavioural motivations in the trust game
- 2016-16 Esther Blanco, Tobias Haller, James M. Walker: Provision of public goods: Unconditional and conditional donations from outsiders
- 2016-15 Achim Zeileis, Christoph Leitner, Kurt Hornik: Predictive bookmaker consensus model for the UEFA Euro 2016
- 2016-14 Martin Halla, Harald Mayr, Gerald J. Pruckner, Pilar García-Gómez: Cutting fertility? The effect of Cesarean deliveries on subsequent fertility and maternal labor supply
- 2016-13 Wolfgang Frimmel, Martin Halla, Rudolf Winter-Ebmer: How does parental divorce affect children's long-term outcomes?
- 2016-12 Michael Kirchler, Stefan Palan: Immaterial and monetary gifts in economic transactions. Evidence from the field
- 2016-11 Michel Philipp, Achim Zeileis, Carolin Strobl: A toolkit for stability assessment of tree-based learners
- 2016-10 Loukas Balafoutas, Brent J. Davis, Matthias Sutter: Affirmative action or just discrimination? A study on the endogenous emergence of quotas forthcoming in Journal of Economic Behavior and Organization
- 2016-09 Loukas Balafoutas, Helena Fornwagner: The limits of guilt

- 2016-08 Markus Dabernig, Georg J. Mayr, Jakob W. Messner, Achim Zeileis: Spatial ensemble post-processing with standardized anomalies
- 2016-07 Reto Stauffer, Jakob W. Messner, Georg J. Mayr, Nikolaus Umlauf, Achim Zeileis: Spatio-temporal precipitation climatology over complex terrain using a censored additive regression model
- 2016-06 Michael Razen, Jürgen Huber, Michael Kirchler: Cash inflow and trading horizon in asset markets
- 2016-05 Ting Wang, Carolin Strobl, Achim Zeileis, Edgar C. Merkle: Scorebased tests of differential item functioning in the two-parameter model
- 2016-04 Jakob W. Messner, Georg J. Mayr, Achim Zeileis: Non-homogeneous boosting for predictor selection in ensemble post-processing
- 2016-03 Dietmar Fehr, Matthias Sutter: Gossip and the efficiency of interactions
- 2016-02 Michael Kirchler, Florian Lindner, Utz Weitzel: Rankings and risktaking in the finance industry
- 2016-01 Sibylle Puntscher, Janette Walde, Gottfried Tappeiner: Do methodical traps lead to wrong development strategies for welfare? A multilevel approach considering heterogeneity across industrialized and developing countries
- 2015-16 Niall Flynn, Christopher Kah, Rudolf Kerschbamer: Vickrey Auction vs BDM: Difference in bidding behaviour and the impact of other-regarding motives
- 2015-15 Christopher Kah, Markus Walzl: Stochastic stability in a learning dynamic with best response to noisy play
- 2015-14 Matthias Siller, Christoph Hauser, Janette Walde, Gottfried Tappeiner: Measuring regional innovation in one dimension: More lost than gained?
- 2015-13 Christoph Hauser, Gottfried Tappeiner, Janette Walde: The roots of regional trust
- 2015-12 Christoph Hauser: Effects of employee social capital on wage satisfaction, job satisfaction and organizational commitment
- 2015-11 **Thomas Stöckl:** Dishonest or professional behavior? Can we tell? A comment on: Cohn et al. 2014, Nature 516, 86-89, "Business culture and dishonesty in the banking industry"
- 2015-10 Marjolein Fokkema, Niels Smits, Achim Zeileis, Torsten Hothorn, Henk Kelderman: Detecting treatment-subgroup interactions in clustered data with generalized linear mixed-effects model trees

- 2015-09 Martin Halla, Gerald Pruckner, Thomas Schober: The cost-effectiveness of developmental screenings: Evidence from a nationwide programme *forthcoming in Journal of Health Economics*
- 2015-08 Lorenz B. Fischer, Michael Pfaffermayr: The more the merrier? Migration and convergence among European regions
- 2015-07 Silvia Angerer, Daniela Glätzle-Rützler, Philipp Lergetporer, Matthias Sutter: Cooperation and discrimination within and across language borders: Evidence from children in a bilingual city forthcoming in European Economic Review
- 2015-06 Martin Geiger, Wolfgang Luhan, Johann Scharler: When do Fiscal Consolidations Lead to Consumption Booms? Lessons from a Laboratory Experiment forthcoming in Journal of Economic Dynamics and Control
- 2015-05 Alice Sanwald, Engelbert Theurl: Out-of-pocket payments in the Austrian healthcare system a distributional analysis
- 2015-04 Rudolf Kerschbamer, Matthias Sutter, Uwe Dulleck: How social preferences shape incentives in (experimental) markets for credence goods forthcoming in Economic Journal
- 2015-03 Kenneth Harttgen, Stefan Lang, Judith Santer: Multilevel modelling of child mortality in Africa
- 2015-02 Helene Roth, Stefan Lang, Helga Wagner: Random intercept selection in structured additive regression models
- 2015-01 Alice Sanwald, Engelbert Theurl: Out-of-pocket expenditures for pharmaceuticals: Lessons from the Austrian household budget survey

University of Innsbruck

Working Papers in Economics and Statistics

2016-22

Christoph Eder

Missing men: World War II casualties and structural change

Abstract

A shock to the sector composition of the local labor market can affect long-run economic development of a location. Because structural change ultimately shifts labor from agriculture to services, an early transition to manufacturing may hamper long-run prosperity. The identification strategy exploits military World War II (WWII) casualties in Austrian municipalities as an exogenous shock to the local labor market. WWII casualties shifted labor out of agriculture into manufacturing in the short-run, which eventually led to a differential path of structural change. In the long-run, I find a strong and robust negative effect of WWII casualties on subsequent economic output.

ISSN 1993-4378 (Print) ISSN 1993-6885 (Online)